Accounting for Mismatch Unemployment

Benedikt Herz European Commission* benedikt.herz@gmail.com

Thijs van Rens University of Warwick and Centre for Macroeconomics J.M.van-Rens@warwick.ac.uk, tvanrens@gmail.com

> This version: January 2019[†] First version: February 2011[‡]

Abstract

We investigate unemployment due to mismatch in the United States over the past three and a half decades. We propose an accounting framework that allows us to estimate the contribution of each of the frictions that generated labor market mismatch. Barriers to job mobility account for the largest part of mismatch unemployment, with a smaller role for barriers to worker mobility. We find little contribution of wage-setting frictions to mismatch.

Keywords: mismatch, structural unemployment, worker mobility, job mobility JEL codes: E24 J61 J62

^{*}The views expressed in this paper are those of the authors and should not be attributed to the European Commission.

[†]We thank the following people for helpful comments and discussions: Árpád Ábrahám, Francesco Amodio, Gadi Barlevy, Régis Barnichon, Björn Brügemann, Tomaz Cajner, Francesca Carapella, Carlos Carrillo Tudela, Vasco Carvalho, Briana Chang, Sekyu Choi, Larry Christiano, Antonio Ciccone, Melvyn Coles, Jim Costain, Steve Davis, Wouter Denhaan, Sijmen Duineveld, Monique Ebell, Bilge Eriş Dereli, Giuliu Fella, Shigeru Fujita, Jordi Galí, Pieter Gautier, Martin Gervais, Albrecht Glitz, Ai Ting Goh, Libertad Gonzalez, Christian Haefke, Giammario Impullitti, Alejandro Justiniano, Philipp Kircher, Marianna Kudlyak, Omar Licandro, Igor Livshits, Rafael Lopes de Melo, Ignacio Lopez, Thomas Lubik, Iourii Manovskii, Alberto Martin, Guido Menzio, Pascal Michaillat, Tomasz Michalski, Claudio Michelacci, Giuseppe Moscarini, Kris Nimark, Nicola Pavoni, Nicolas Petrosky-Nadeau, Giacomo Ponzetto, Giorgio Primiceri, Roland Rathelot, Richard Rogerson, Ayşegül Şahin, Kurt Schmidheiny, Petr Sedláček, Matthew Shapiro, Rob Shimer, Henry Siu, Eric Smith, Andrea Tambalotti, Giorgio Topa, Lawrence Uren, Jaume Ventura, Gianluca Violante, Ludo Visschers, Etienne Wasmer, Sweder van Wijnbergen and our discussants Winfried Koeniger and Michael Reiter. Thijs van Rens gratefully acknowledges the hospitality of the UNSW Sydney School of Economics during the final stages of preparing this paper.

[‡]Early versions of this paper were circulated under the title "Structural Unemployment".

1 Introduction

After the end of the Great Recession in December 2007, unemployment in the United States remained high for more than half a decade. One explanation that was suggested is a mismatch in the skills or geographic location of the available jobs and workers, a view that seemed to be supported by a decline in aggregate matching efficiency (Elsby, Hobijn, and Şahin (2010), Barnichon and Figura (2010)) and geographic mobility (Frey (2009)). Direct estimates using disaggregated data confirm that mismatch rose in the recession (Şahin, Song, Topa, and Violante (2014)). However, the literature offers little insight into the reasons for this increase.

In this paper, we estimate mismatch unemployment on the U.S. labor market from 1979 to 2015, and we explore what frictions caused mismatch to arise. To do so, we use a model to derive an accounting framework that puts just enough structure on the data to allow us to quantify the sources of mismatch unemployment.

The labor market in our model consists of multiple submarkets or segments. Mismatch is defined as inefficient dispersion in labor market conditions, in particular the job-finding rate, across labor market segments. Within segments, frictions prevent the instantaneous matching of unemployed workers to vacant jobs, resulting in search unemployment in the tradition of Diamond (1982), Mortensen (1982) and Pissarides (1985). Across segments, frictions generate dispersion in labor market conditions, which gives rise to mismatch unemployment.¹

Three types of frictions generate mismatch: worker-mobility costs, job-mobility costs, and wage-setting frictions. Worker-mobility frictions prevent an unemployed worker in one submarket from taking up a job in a different submarket. For example, if we think of labor market segments as occupations, worker-mobility frictions might prevent an unemployed construction worker from finding a job as a retail sales representative, a displaced steelworker from filling a vacancy for a nurse, or an unemployed engineer to take up a job as a lawyer. In these examples, worker mobility frictions take the form of training costs and occupational licensing. If we think of labor market segments as states, regions or commuting zones, frictions may also take the form of moving costs, etc. Job-mobility frictions prevent firms from substituting positions (e.g., onsite technicians) in submarkets where labor is scarce with different positions (e.g., online customer support representatives) for which the supply of workers is more abundant. We think of

¹The definition of mismatch as a deviation from an efficient allocation follows the recent empirical literature, in particular Şahin, Song, Topa, and Violante (2014). Although not directly related, this definition is also consistent with the theoretical literature. Shimer (2007) shows that mismatch between the distributions of workers and jobs over segments of the labor market gives rise to a relationship between the job-finding probability and labor-market tightness that is very similar to the relationship obtained if there are search frictions and an aggregate matching function. Stock-flow matching, as in Coles, Jones, and Smith (2010); rest unemployment, as in Alvarez and Shimer (2011); reallocation unemployment as in Carrillo-Tudela and Visschers (2013), Wong (2012) or Chang (2011); waiting unemployment as in Birchenall (2011); mismatch unemployment as in Wiczer (2013); and move unemployment as in Pilossoph (2014) are all closely related to this concept of unemployment due to mismatch. As opposed to these studies, the focus of our paper is empirical. One way to think about the contribution of this paper is to provide a set of facts that can be used to test the theoretical models of mismatch unemployment.

these frictions as rigidities in the production technology, or as barriers to entry in particular occupations or industries. Both of these frictions may explain why a situation of mismatch between the distribution of vacancies and unemployed workers over submarkets can persist. Wage-setting frictions prevent the wage from reflecting the relative abundance or shortage of workers in different submarkets. This type of friction, which we could think of as a type of wage rigidity, generates mismatch because (everything else equal) workers will prefer to look for jobs in submarkets where wages are relatively high, whereas firms are more inclined to try and recruit in submarkets where wages are relatively low. As an example, this type of friction might explain why many workers with engineering backgrounds work in finance, while manufacturing firms find recruiting engineers difficult.

Our approach to estimate the sources of mismatch unemployment uses data on joband worker-finding rates, and worker and job surplus by labor market segments, which we operationalize as occupations, states, or industries. We construct these variables over the 1979-2015 period from the Current Population Survey (CPS) and the National Income and Product Accounts (NIPA).

We argue that mismatch is an important reason for unemployment. Our estimates show that mismatch across detailed occupations and states is responsible for about a fifth of fluctuations in, and for around 13% of the level of unemployment. The cyclical behavior of mismatch unemployment is very similar to that of the overall unemployment rate. This finding is driven by the fact that dispersion in labor market conditions across states and industries moves closely with the business cycle, similar to what Abraham and Katz (1986) documented over three decades ago.² The unemployment that derives from this dispersion is as cyclical as the overall unemployment rate, and no more persistent. As a corollary, the nature of the increase in unemployment in the Great Recession was no different from the increase in previous recessions, although it was, of course, more severe.³ The absence of a secular trend in mismatch unemployment indicates that the increase in mismatch unemployment was not "structural", in the sense that it would not respond to stabilization policy.⁴

²In response to the "structural shifts view" of recessions put forward by Lilien (1982), which holds that recessions are periods of reallocation between industries, Abraham and Katz showed that aggregate shocks can give rise to countercyclical fluctuations in dispersion of employment growth across sectors.

³This result is not inconsistent with the observation that there was an outward shift in the Beveridge curve, the negatively sloped relationship between vacancies and unemployment, which indicates a decline in aggregate matching efficiency and provides much of the basis for the argument that there was an unprecedented increase in mismatch in the Great Recession (Elsby, Hobijn, and Şahin (2010), Lubik (2013)). While an increase in mismatch indeed reduces matching efficiency (Shimer (2007)), there are many other causes for shifts in the Beveridge curve as well, including changes in the separation rate and demographics. Controlling for these factors, the remaining role for mismatch is very small (Barnichon and Figura (2010)). For the same reason, our findings do not contradict the observation that exogenous shocks to mismatch are not an important as a source of unemployment fluctuations (Furlanetto and Groshenny (2016)).

⁴In the wake of the Great Recession, this was a widely held view, advocated most prominently by Narayana Kocherlakota (2010), the president of the Federal Reserve Bank of Minneapolis, who argued that "it is hard to see how the Fed can do much to cure this problem. Monetary stimulus has provided conditions so that manufacturing plants want to hire new workers. But the Fed does not have a means to transform construction workers into manufacturing workers." See Estevão and Tsounta (2011) and

Our most interesting and novel set of results concerns the sources of labor market mismatch. We find that almost all mismatch is caused by barriers to worker and job mobility, and that the latter are much more important than the former. Little or no mismatch is due to wage-setting frictions. These conclusions are based on testing the strong predictions generated by our framework for the patterns we should observe in the data in the absence of the various frictions that can give rise to mismatch. In particular, if there are no barriers to worker mobility, a no-arbitrage condition dictates that we should see a negative correlation between wages (measuring how attractive it is to have a job in a given state or industry) and job-finding rates (how hard it is to find these jobs). We indeed find this correlation in the data.

The early empirical literature on mismatch focused on shifts in the Beveridge curve, trying to use aggregate data to estimate matching efficiency (Lipsey (1965), Abraham (1987), Blanchard and Diamond (1989), Barnichon and Figura (2010)). Two more recent contributions use disaggregated data and are closely related to this paper. Şahin, Song, Topa, and Violante (2014) use data on unemployment and vacancies by counties, occupations, and industries (from the JOLTS and the HWOL data for the 2001-2011 and 2005-2011 periods, respectively), to construct indices of mismatch. Barnichon and Figura (2015) use the CPS to explore how much dispersion in labor market conditions contributes to movements in matching efficiency.⁵ The contribution of this paper is the accounting framework that allows us to decompose mismatch into its sources, and to estimate the contribution of each of these sources to unemployment.

This paper is organized as follows: In the next section we define mismatch as any deviation from the allocation chosen by a planner who can freely allocate vacancies and unemployed workers across submarkets of the labor market; we also set up the model environment to implement this definition. In Section 3, we show how the competitive equilibrium of this model can be used to decompose mismatch into the different sources of inefficient dispersion in job-finding rates. We identify three sources of mismatch: worker-mobility costs, job-mobility costs and wage-setting frictions. Section 4 presents the details of our proposed "mismatch accounting" procedure and discusses some important caveats. Section 5 describes the data used in the estimation, and explains in detail how we construct the empirical counterparts of the variables that define a labor market segment in our model. Finally, Section 6 presents the empirical results and Section 7 concludes.

Groshen and Potter (2003) for versions of this argument. Early critics include Krugman (2010), DeLong (2010), P. and Spletzer (2012), and Peter Diamond (2011), who notes in his Nobel lecture that "there is a long history of claims that the latest technological or structural developments make for a new, long-term, high level of unemployment, but these have repeatedly been proven wrong." (p.1065). Kocherlakota later changed his views in light of the evidence (New York Times (2014)).

⁵Another related recent paper is Herz (2017), who examines the role of skill mismatch for unemployment using micro-data on displaced workers. If no vacancies matching a displaced worker's skill set are currently available, she can either fill a vacancy for a job she is not trained for, and suffer a wage loss as a result; or she can wait until until a more suitable vacancy opens up. Herz empirically examines this trade-off using a difference-in-difference approach, and finds that such "wait unemployment" accounts for a substantial part of aggregate unemployment.

2 Mismatch

We define mismatch, following Şahin, Song, Topa, and Violante (2014), as any deviation from the allocation of unemployed workers and vacancies over labor market segments that a social planner would choose. Mismatch unemployment is unemployment that arises due to this mismatch, i.e., the difference between actual unemployment and unemployment in the social planner allocation. We show that, under some conditions, a social planner would equalize job-finding probabilities across labor market segments. Under these conditions, mismatch unemployment can be equivalently defined as unemployment due to dispersion in job-finding rates.

2.1 Model Environment

Consider a labor market that consists of segments indicated by i. The idea is that each unemployed worker cannot match with each vacancy. A segment, or submarket, is defined as the subset of jobs that a given unemployed worker searches for, or the subset of unemployed workers that can form a match with a given vacancy.

There are n_{it} workers employed in segment *i* at time *t*, who produce output using a production technology that requires only labor $f(n_{it}; z_{it})$, where z_{it} is a production efficiency shifter, which may vary across segments. Each labor market segment is subject to frictions in the tradition of Diamond (1982), Mortensen (1982) and Pissarides (1985). Job matches are formed from u_{it} unemployed workers and v_{it} vacancies using a constant-returns-to-scale matching technology $m(u_{it}, v_{it}; \phi_{it}) = \phi_{it} u_{it}^{\mu} v_{it}^{1-\mu}$, with $0 < \mu < 1$ and where ϕ_{it} is a matching efficiency shifter, which determines the job-finding and vacancy-filling probabilities $p_{it} = m(u_{it}, v_{it}; \phi_{it}) / u_{it} = \phi_{it} (v_{it}/u_{it})^{1-\mu}$ and $q_{it} = m(u_{it}, v_{it}; \phi_{it}) / v_{it} = \phi_{it} (v_{it}/u_{it})^{-\mu}$. Matches are destroyed with an exogenous probability δ_i in each period, so that employment in each segment evolves as $n_{it+1} = (1 - \delta_i) n_{it} + m(u_{it}, v_{it}; \phi_{it})$.

Unemployed workers engage in home production in the amount of b_{it} and workers not participating in the labor force produce λ^u , which includes any unemployment and welfare benefits and utility from leisure. Vacancies cost $g(v_{it}; \kappa_{it})$ in each period, where κ_{it} is a vacancy cost shifter, and there is an additional opportunity cost λ^v of keeping a position open, whether filled or unfilled, which we may think of as the revenue from renting out the capital associated with the position if it is closed (set $\lambda^v = 0$ to represent free entry of vacancies). The segment-specific shocks z_{it} , ϕ_{it} , b_{it} and κ_{it} all follow exogenous Markov processes. Workers have linear utility over consumption only,⁶ and discount future periods at rate $\beta = 1/(1 + r)$.

⁶For simplicity. The efficient and equilibrium allocations are unchanged if we assume instead that households consisting of many workers share risk between their members. Utility from leisure is included in home production b_{it} and λ^u .

2.2 Efficient Allocation of Unemployed and Vacancies

The social planner allocates unemployed workers and vacancies over labor market segments to maximize the expected net present value of utility of the representative worker, which is equivalent to maximizing the expected net present value of output net of vacancy costs $\sum_{i} f(n_{it}; z_{it}) + b_{it}u_{it} - g(v_{it}; \kappa_{it}) + \lambda^u (1 - n_{it} - u_{it}) + \lambda^v (1 - n_{it} - v_{it})$, subject to the evolution of employment in each segment under the matching technology.

In appendix A.1, we show that the social planner chooses an allocation that satisfies the following efficiency condition,

$$p_{it} = \phi_{it} \left(\frac{1-\mu}{\mu} \frac{\lambda^u - b_{it}}{\lambda^v + g'\left(v_{it}; \kappa_{it}\right)} \right)^{1-\mu} \tag{1}$$

where $p_{it} = \phi_{it} (v_{it}/u_{it})^{1-\mu}$ is the job-finding probability in segment *i* at time *t*, and μ is the unemployment share parameter in the matching function, $m(u_{it}, v_{it}; \phi_{it}) = \phi_{it} u_{it}^{\mu} v_{it}^{1-\mu}$.

A version of efficiency condition (1) may be found in Şahin, Song, Topa, and Violante (2014), who assume the production function is linear but that vacancy posting costs increase with the amount of vacancies in a segment. If we set $f(n_{it}; z_{it}) = z_{it}n_{it}$ and $g(v_{it}; \kappa_{it}) = (1/(1+\varepsilon)) \kappa_{it}^{\varepsilon} v_{it}^{1+\varepsilon}$, then (1) reduces to (A36) in Şahin, Song, Topa, and Violante (2014), see appendix A.1.⁷

2.3 Dispersion in Labor Market Conditions and Mismatch

If home production, vacancy costs and matching efficiency are homogeneous, $b_{it} = b_t$, $g'(v_{it}; \kappa_{it}) = \kappa_t$ and $\phi_{it} = \phi_t$, then condition (1) prescribes that in the efficient allocation the job-finding probability must be equal in all labor market segments, $p_{it} = \bar{p}_t$. Furthermore, the planner allocates more vacancies and unemployed workers to segments where productivity is high, so that the expected net present value of the marginal product of labor is equal as well, see appendix A.1.4.⁸ This is our benchmark allocation with full equalization of labor market conditions.⁹

$$m_u \left(u_{it}, v_{it}; \phi_{it} \right) S_{it} = \lambda^u - b_{it} \tag{2}$$

⁷If the vacancy distribution is exogenous (not under the control of the planner), as in the baseline of Şahin, Song, Topa, and Violante (2014), condition (1) is replaced by,

where S_{it} is the expected net present value of the marginal productivity of labor $f'(n_{it}; z_{it})$, which equals $(z_{it} - \lambda^u - \lambda^v) / (r + \delta_i)$ if we further assume that the production function is linear and productivity z_{it} follows a random walk (derivation in appendix A.1). This is condition (2) in Şahin, Song, Topa, and Violante (2014).

⁸We need decreasing returns in production function $f(n_{it}; z_{it})$ for this result. This is a reasonable assumption if production requires not only labor but also capital, which may be harder to adjust than employment.

⁹If vacancies are exogenous, but in addition to home productivity $b_{it} = b_t$ and matching efficiency $\phi_{it} = \phi_t$, market productivity and separation rates are homogeneous as well, $z_{it} = z_t$ and $\delta_i = \delta$ so that $S_{it} = S_t$, then (2) states that in the efficient allocation the marginal contribution of unemployment to job matches $m_{u,it}(u_{it}, v_{it}; \phi_{it})$ must be equalized. In this case, which is condition (1) in Şahin, Song, Topa, and Violante (2014), the job-finding probability is equalized as well.

There are two reasons for dispersion in labor market conditions: parameter heterogeneity and mismatch. Heterogeneity in the parameters b_{it} (productivity of home production of the unemployed), κ_{it} (vacancy maintenance costs), and ϕ_{it} (matching efficiency), generates efficient dispersion in job-finding rates. In the empirical part of this paper, we try to control for heterogeneity in these parameters, as described in Section 4.2 below. Mismatch is defined as deviations from condition (1), i.e., *inefficient* deviations from the benchmark of fully equalized labor market conditions.

3 Sources of Mismatch

In this Section, we solve for an equilibrium of the model described in Section 2.1 above, in which there are no frictions other than the search friction as in Diamond (1982), Mortensen (1982) and Pissarides (1985). We will show that the allocation of unemployed workers and vacancies in this equilibrium is the same as the efficient allocation and therefore refer to this equilibrium as the no-mismatch equilibrium. The equilibrium conditions can be used to decompose mismatch, i.e., deviations from the efficient no-mismatch equilibrium allocation, into the contribution of three different types of frictions: barriers to the mobility of workers between labor market segments, barriers to the mobility of vacancies between segments, and inefficient wage dispersion. Below, we describe the equilibrium conditions in words. The derivations may be found in appendix A.2.

3.1 Worker Mobility

If unemployed workers are free to move between labor market segments, then they must be indifferent about the labor market in which they will search for a job; this is described by the following equilibrium condition, see appendix A.2.1, which we call the "worker-mobility condition."

$$p_{it}S^W_{it} = \lambda^u - b_{it} \tag{3}$$

 S_{it}^W is the surplus that is expected to realize for a worker who is allocated to segment *i* if she finds a job there in period *t*, i.e., the expected net present value of wages w_{it} in that segment. If wages follow a random walk, then $S_{it}^W = (w_{it} - \lambda^u) / (r + \delta_i)$ is the value of an infinite stream of (expected) wages w_{it} , net of home production of non-participants, discounted by the rate of time preference and the probability δ_i that the worker loses the job again.

Intuitively, worker-mobility condition (3) is a no-arbitrage condition. In the absence of parameter heterogeneity, $b_{it} = b_t$, it states that attractive jobs must be hard to find, and unattractive jobs easy to find.¹⁰ If home productivity b_{it} differs across segments,

¹⁰The insight is the same as that of the Harris and Todaro (1970) model of rural-urban migration. In the context of worker mobility, it should not be surprising that some (urban) areas have much lower job-finding rates (higher unemployment) if wages are much higher there. Similarly, Montgomery (1991) proposes differences in job-finding rates as an explanation for persistent wage differentials across

then segments with high home productivity must have low job-finding rates or low worker surplus, or both.

3.2 Job Mobility

If firms can freely relocate vacancies across labor market segments, a no-arbitrage condition holds that is similar to the worker-mobility condition above. We call this condition the "job-mobility condition." Let $q_{it} = m(u_{it}, v_{it}; \phi_t) / v_{it}$ be the probability that a firm fills a vacancy in segment *i* in period *t*. Then, as derived in appendix A.2.2,

$$q_{it}S_{it}^F = \lambda^v + g'\left(v_{it};\kappa_{it}\right) \tag{4}$$

where S_{it}^F is the surplus a firm realizes when its vacancy in segment *i* is filled in period *t*, i.e. the expected net present value of profits. If profits follow a random walk, then $S_{it}^F = (f'(n_{it}; z_{it}) - w_{it} - \lambda^v) / (r + \delta_i).$

In the absence of heterogeneity, e.g., if $g'(v_{it}; \kappa_{it}) = \kappa_{it} = \kappa_t$, vacancies in segments where jobs are expected to generate large profits must be hard to fill, and vacancies in low-profit segments must be easy to fill. With heterogeneity, vacancies in segments with high vacancy-maintenance costs must be easy to fill or expected to generate high profits or both. As in the social planner problem, the opportunity cost of creating more vacancies in total λ^v equals zero if the firm is able to freely generate vacancies (i.e., if there is "free entry"). This does not change any of the predictions of the model for dispersion across segments.

3.3 Matching Technology

The job-finding probability p_{it} in worker-mobility condition (3) and the vacancy-filling probability q_{it} in job-mobility condition (4) are related to each other, because both depend on the vacancy-unemployment ratio through matching technology. For our Cobb-Douglas matching function, $m(u_{it}, v_{it}; \phi_{it}) = \phi_{it} u_{it}^{\mu} v_{it}^{1-\mu}$, we get that $p_{it} = \phi_t (v_{it}/u_{it})^{1-\mu}$ and $q_{it} = \phi_t (v_{it}/u_{it})^{-\mu}$, which implies

$$p_{it}^{\mu} q_{it}^{1-\mu} = \phi_{it} \tag{5}$$

Different from the optimality conditions (3) and (4), equilibrium condition (5) is a technological constraint, which plays the role of the market-clearing condition on a perfectly competitive labor market in connecting labor supply with labor demand.

In the absence of parameter heterogeneity, $\phi_{it} = \phi_t$, the job-finding probability and the job-filling probability are inversely related to each other. With heterogeneity, both probabilities are higher in segments with high matching efficiency, and lower in segments with low matching efficiency. If the elasticity of the matching function μ is not constant

industries.

across segments, then condition (5) still holds in first order approximation. In this case, dispersion around the condition reflects heterogeneity in ϕ_{it} as well as μ_{it} .

3.4 Wage Determination

Worker mobility depends on the surplus of a job that accrues to the worker S_{it}^W , see (3). Job mobility depends on the surplus that accrues to the firm S_{it}^F , see (4). Thus, to close the model we need to specify how the total surplus of a match is divided between worker and firm. In the no-mismatch equilibrium, we assume the wage is set such that the worker gets a share μ of total match surplus, where μ is the unemployment share parameter in the matching technology.

$$\frac{S_{it}^W}{\mu} = \frac{S_{it}^F}{1-\mu} \tag{6}$$

This surplus-sharing rule can be justified as the outcome of a Nash bargaining process, where the bargaining power of workers satisfies the Hosios (1990) condition. Under this wage-setting condition, the equilibrium is efficient, see below equation (11) in Section 3.5.

3.5 No-Mismatch Equilibrium

The equilibrium conditions are all log-linear in the endogenous variables of the model p_{it} , q_{it} , S_{it}^W and S_{it}^F . It is convenient, therefore, to write the equilibrium conditions in logs, using a tilde over a variable to denote its logarithm, $\tilde{X}_{it} = \log X_{it}$. Taking logs of conditions (3), (4), (5) and (6), we can summarize the equilibrium with the following four conditions,

$$\tilde{p}_{it} + \tilde{S}_{it}^W = \alpha_{it}^{WM} \tag{7}$$

$$\tilde{q}_{it} + \tilde{S}_{it}^F = \alpha_{it}^{JM} \tag{8}$$

$$\frac{\mu}{1-\mu}\tilde{p}_{it} + \tilde{q}_{it} = \alpha_{it}^{MT} \tag{9}$$

$$\tilde{S}_{it}^W - \tilde{S}_{it}^F = \alpha^{WD} \tag{10}$$

where $\alpha_{it}^{WM} = \log(\lambda^u - b_{it}), \ \alpha_{it}^{JM} = \log(\lambda^v + g'(v_{it}; \kappa_{it})), \ \alpha_{it}^{MT} = (1/(1-\mu))\log\phi_{it}$ and $\alpha^{WD} = \log(\mu/(1-\mu))$ so that heterogeneity in $\alpha_{it}^{WM}, \ \alpha_{it}^{JM}$ and α_{it}^{MT} reflects heterogeneity in home productivity, vacancy costs and matching efficiency, respectively.

Solving this system of equations, we get an explicit expression for segment-specific job finding probabilities in equilibrium.

$$\tilde{p}_{it} = (1 - \mu) \left(\alpha_{it}^{WM} - \alpha_{it}^{JM} + \alpha_{it}^{MT} - \alpha^{WD} \right)$$
(11)

Substituting the definitions of α_{it}^{WM} , α_{it}^{JM} and α_{it}^{MT} and α^{WD} into (11), it is straightforward to show that this equilibrium distribution of job-finding probabilities equals the efficient distribution as in (1).

As in the efficient allocation, the equilibrium job-finding probabilities are constant across labor market segments, even if productivity z_{it} , wages w_{it} , and separation probabilities δ_i vary across submarkets, if the crucial parameters home productivity, vacancy costs and matching efficiency are homogeneous, i.e., if $b_{it} = b_t$, $g'(v_{it}; \kappa_{it}) = \kappa_t$ and $\phi_{it} = \phi_t$ so that $\alpha_{it}^{WM} = \alpha_t^{WM}$, $\alpha_{it}^{JM} = \alpha_t^{JM}$ and $\alpha_{it}^{MT} = \alpha_t^{MT}$. It is straightforward to show that in this case the vacancy-filling probability q_{it} , and worker and firm surplus S_{it}^W and S_{it}^F are equalized as well, and the model effectively reduces to a standard search model with a single, unsegmented labor market.

3.6 Discussion

The no-mismatch equilibrium conditions for worker and job mobility can be interpreted in various ways. Our preferred interpretation is as no-arbitrage conditions, because that interpretation allows us to posit the conditions with very few assumptions. However, we also showed how these conditions can be derived as optimality conditions of, respectively, a household and a firm in a model of a segmented labor market subject to search frictions within each segment. Very similar conditions could be derived in the context of a directed search model as well.

The assumptions we need to derive conditions (7), (8), (9) and (10) are relatively uncontroversial, so that the framework so far is quite general (we will need to make many more assumptions to operationalize the procedure, which we discuss in the next section). One assumption in particular, which does not affect the equilibrium conditions, is worth mentioning explicitly. We assume that workers and firms can only search in one labor market segment at the same moment in time. The conditions would be unchanged if we relax this assumption and assume that workers and firms can distribute search effort over multiple segments, as long as the total amount of search effort is finite, so that more intensive search in one segment comes at the cost of reduced search intensity in another segment. However, in this case our approach will overstate the effect of deviations from the worker mobility conditions for unemployment, as pointed out by Marinescu and Rathelot (2016). We return to this issue when we discuss the robustness of our results in Section 6.4.

In the next section, we discuss our procedure to decompose mismatch unemployment into its sources using the no-mismatch equilibrium conditions. We want to emphasize that we are not taking a stance ex-ante on whether or not we expect these conditions to be satisfied in the data. These no-mismatch equilibrium conditions are just benchmark conditions; that is, they are conditions, under which labor market conditions are fully equalized across segments. Deviations from the benchmark conditions represent sources of labor market mismatch.

4 Mismatch Accounting

In general, the distribution of job-finding probabilities across labor market segments may deviate from the efficient equilibrium, generating mismatch. The equilibrium conditions allow us to decompose any deviations from the no-mismatch allocation into deviations from optimality condition (7) for workers, deviations from optimality condition (8) for firms, or deviations from the efficient surplus-sharing rule (10). We refer to these three sources of mismatch as worker-mobility frictions, job-mobility frictions, and wage-setting frictions, and to the decomposition exercise as mismatch accounting. Since condition (9) is a technological constraint rather than a behavioral equation, we do not anticipate deviations from this condition other than efficient deviations due to heterogeneity in matching efficiency.

In this section we describe the implementation of our mismatch accounting procedure. First, we show why it is not possible to identify mismatch from efficient dispersion in labor market conditions due to parameter heterogeneity without further assumptions. Then, we describe how we control for parameter heterogeneity by assuming it is timeinvariant, similar to controlling for fixed effects in a regression. Third, we discuss how we can summarize the aggregate effects of labor market mismatch as mismatch unemployment, unemployment that arises due to mismatch or the difference between actual unemployment and the unemployment rate that would prevail in the social planner allocation. Finally, we put everything together and then discuss a few caveats and limitations.

4.1 Frictions and Wedges

If there are other frictions on the labor market than just within-segment search frictions, then there is no reason why the no-mismatch equilibrium conditions (7), (8) and (10) should be satisfied. We can therefore represent such frictions in terms of the deviations from these conditions observed in the data. We denote these deviations, which we call wedges following Chari, Kehoe, and McGrattan (2007), by γ_{it}^{WM} , γ_{it}^{JM} and γ_{it}^{WD} . Then, the equilibrium allocation observed in the data can be written as

$$\tilde{p}_{it} + \tilde{S}_{it}^W = \alpha_{it}^{WM} + \gamma_{it}^{WM} \tag{12}$$

$$\tilde{q}_{it} + \tilde{S}_{it}^F = \alpha_{it}^{JM} + \gamma_{it}^{JM} \tag{13}$$

$$\frac{\mu}{1-\mu}\tilde{p}_{it} + \tilde{q}_{it} = \alpha_{it}^{MT} \tag{14}$$

$$\tilde{S}_{it}^W - \tilde{S}_{it}^F = \alpha^{WD} + \gamma_{it}^{WD} \tag{15}$$

and the job finding rates across segments are

$$\tilde{p}_{it} = (1-\mu) \left(\alpha_{it}^{WM} - \alpha_{it}^{JM} + \alpha_{it}^{MT} - \alpha^{WD} + \gamma_{it}^{WM} - \gamma_{it}^{JM} - \gamma_{it}^{WD} \right)$$
(16)

The basic idea behind our proposed exercise of mismatch accounting is to measure the wedges γ_{it}^{WM} , γ_{it}^{JM} and γ_{it}^{WD} from equations (12), (13) and (15), and then use equation (16) to quantify the contribution of each of the three wedges to dispersion in job-finding rates. However, since not only wedges γ_{it}^{WM} , γ_{it}^{JM} and γ_{it}^{WD} , but also parameter heterogeneity, as measured by α_{it}^{WM} , α_{it}^{JM} and α_{it}^{MT} , generate dispersion in the observables \tilde{p}_{it} , \tilde{q}_{it} , \tilde{S}_{it}^{W} and \tilde{S}_{it}^{F} , it is not possible to implement this procedure without more data or further assumptions on the parameter heterogeneity.

4.2 Heterogeneity

Our identifying assumption is that parameter heterogeneity is constant over time, i.e., $b_{it} = b_i$, $\kappa_{it} = \kappa_i$ and $\phi_{it} = \phi_i$, so that $\alpha_{it}^{WM} = \alpha_i^{WM}$, $\alpha_{it}^{JM} = \alpha_i^{JM}$ and $\alpha_{it}^{MT} = \alpha_i^{MT}$, Under this assumption, we can control for parameter heterogeneity in a way that is similar in spirit to controlling for fixed effects in a regression: by removing the segmentspecific time-series averages from our data.

Let a hat over a variable denote the deviation of its logarithm from the segmentspecific time-series average, i.e. $\hat{X}_{it} = \tilde{X}_{it} - \frac{1}{T} \sum_{t=1}^{T} \tilde{X}_{it}$, where T is the final time period in the sample. Then, taking deviations from equations (12), (13), (14) and (15), the allocation observed in the data can be written as,

$$\hat{p}_{it} + \hat{S}_{it}^W = \hat{\gamma}_{it}^{WM} \tag{17}$$

$$\hat{q}_{it} + \hat{S}_{it}^F = \hat{\gamma}_{it}^{JM} \tag{18}$$

$$\frac{\mu}{1-\mu}\hat{p}_{it} + \hat{q}_{it} = 0 \tag{19}$$

$$\hat{S}_{it}^W - \hat{S}_{it}^F = \hat{\gamma}_{it}^{WD} \tag{20}$$

because $\hat{\alpha}_i^{WM} = \hat{\alpha}_i^{JM} = \hat{\alpha}_i^{MT} = \hat{\alpha}^{WD} = 0$, where $\hat{\alpha}_i^{XX} = \alpha_i^{XX} - \frac{1}{T} \sum_{t=1}^T \alpha_i^{XX}$ and $\hat{\gamma}_{it}^{XX} = \gamma_{it}^{XX} - \frac{1}{T} \sum_{t=1}^T \gamma_{it}^{XX}$.

Solving this system of equations, we get

$$\hat{p}_{it} = (1-\mu) \left(\hat{\gamma}_{it}^{WM} - \hat{\gamma}_{it}^{JM} - \hat{\gamma}_{it}^{WD} \right)$$
(21)

Under our identifying assumption that parameter heterogeneity is constant over time, all observed dispersion in segment-specific job-finding rates in deviations from their time-series averages \hat{p}_{it} is due to dispersion in the wedges, i.e. to mismatch.

We treat parameter heterogeneity as unobservable, because it is not feasible to convincingly measure heterogeneity in the parameters of our model. Although some data on these parameters is available, it is likely to be of lower quality than data on p_{it} , q_{it} , S_{it}^W and S_{it}^F . For example, while there is information on unemployment benefits across states, which is informative about dispersion in b_{it} , we do not have information on unemployment benefits across industries, nor do we have estimates for the dispersion in the value of leisure across either states or industries. To the best of our knowledge, there is no data at all on vacancy maintenance costs or matching efficiency by states or industries. Our solution is to assume that parameter heterogeneity is time invariant.

Our approach to deal with heterogeneity has advantages as well as disadvantages. The most obvious disadvantage is that we do not control for time variation in parameter heterogeneity. It seems reasonable to assume that there is more variation in parameters across states or industries than within segments over time.¹¹ Nevertheless, if there is heterogeneous time variation in unemployment benefits, vacancy maintenance costs or matching efficiency, then we will spuriously attribute the dispersion in labor market conditions that arises because of it to mismatch. In Section 4.4, we discuss to what extent this bias may affect our results.

A second disadvantage is that, in addition to dispersion due to heterogeneity, we may also remove dispersion due to mismatch. If differences across submarkets due to worker-mobility, job-mobility or wage setting frictions are persistent over time, then some of the dispersion in γ_{it}^{WM} , γ_{it}^{JM} and γ_{it}^{WD} will be removed if we take deviations from time-series averages. In that sense, our estimates should be viewed as a lower bound for the contribution of mismatch to aggregate job finding and unemployment.

Offsetting these disadvantages is an important advantage of our approach: in addition to parameter heterogeneity, the approach also controls for all other time-invariant heterogeneity, observable as well as unobservable, and across workers as well as across firms. Heterogeneity across workers and firms is a concern because we estimate the contribution of mismatch to unemployment from the dispersion in wages, profits, and finding probabilities. Heterogeneity generates dispersion that is unrelated to mismatch. Wages, profits and even job-finding rates may vary across workers not only because of deviations from the no-mismatch equilibrium conditions, but also because workers have different education, experience, and other characteristics. Failing to control for these differences would spuriously attribute the dispersion across labor market segments they generate to mismatch.¹² Residual wage differentials are due at least in part to compensating differentials: non-monetary job amenities such as flexible hours or safe working conditions, in return for which workers are willing to accept lower wages, see Rosen (1979) and Roback (1982).¹³ These differences, which are completely unobservable in our dataset, also generate dispersion that is not inefficient, and, thus, should not be attributed to mismatch.

¹¹For unemployment benefits across states, for which data are available, we find that 77% of the variance in the UI replacement ratio is due to variation between states, and the between-state standard deviation of 0.044 is close to the overall standard deviation of 0.050. Source: UI replacement ratios (definition 1) by state and year 1997-2017 from the US Department of Labor, available at https://oui.doleta.gov/unemploy/ui_replacement_rates.asp.

¹²In Section 6.4, we show that explicitly controlling for observable worker heterogeneity affects the results very little, indicating that this heterogeneity is already adequately controlled for in our baseline results.

¹³One of these compensating differentials is explicitly taken into account in our calculations, which is the separation probability. However, this is only one of many differences between jobs.

4.3 Effect of Mismatch on Unemployment

Mismatch unemployment is given by the difference between the actual unemployment rate and the counterfactual unemployment that would prevail if all wedges were zero. The wedges generate dispersion in labor market conditions over and above the counterfactual dispersion in the no-mismatch equilibrium. But how does dispersion in labor market conditions affect the aggregate unemployment rate?

The aggregate unemployment rate is determined by the aggregate separation rate δ_t and the aggregate job-finding probability \bar{p}_t . Using the approximation that the aggregate unemployment rate is in steady state given these worker flows, and taking a first-order Taylor approximation, we can show that the effect of mismatch on unemployment is roughly proportional to its effect on the job-finding rate,

$$\bar{u}_t = \frac{\bar{\delta}_t}{\bar{\delta}_t + \bar{p}_t} \Rightarrow \log \bar{u}_t - \log \bar{u}_t^* \simeq \frac{\bar{p}_t}{\bar{\delta}_t + \bar{p}_t} \left(\log \bar{p}_t^* - \log \bar{p}_t\right) \tag{22}$$

where \bar{u}_t^* and \bar{p}_t^* denote the counterfactual unemloyment and job-finding rate in the absence of mismatch. The effect of mismatch on unemployment through the aggregate separation rate is negligible compared to the effect through the aggregate job-finding rate for two reasons. First, the first-order effect through $\log \bar{\delta}_t - \log \bar{\delta}_t^*$ is multiplied by $\bar{\delta}_t / (\bar{\delta}_t + \bar{p}_t)$, which is much smaller than the $\bar{p}_t / (\bar{\delta}_t + \bar{p}_t)$ factor multiplying the effect through the job-finding rate because $\bar{\delta}_t << \bar{p}_t$ in the US data. Second, the aggregate separation rate $\bar{\delta}_t = \sum_i \delta_i n_{it} / \sum_i n_{it}$ is not affected by reallocating unemployed workers and vacancies in the short run, and the long-run effect through changes in the steady-state distribution of employment n_{it} across segments is small compared to the change in the aggregate job-finding rate.¹⁴

The counterfactual job-finding rate in the absence of mismatch is higher than the actual job-finding rate, $\bar{p}_t^* > \bar{p}_t$, because job-finding rates are concave in vacancyunemployment ratios, $p_{it} = \phi_t \theta_{it}^{1-\mu}$ with $0 < \mu < 1$, so that dispersion in labor market conditions lowers the average job-finding probability. To get a closed-form expression for the aggregate effect of mismatch, we assume that the distribution of p_{it} is log-normal. Under this assumption, the relative contribution of mismatch to the aggregate jobfinding rate is proportional to the dispersion in job-finding rates across labor market segments,

$$\log \bar{p}_t^* - \log \bar{p}_t = \frac{1}{2} \frac{\mu}{1-\mu} V\left[\hat{p}_{it}\right]$$
(23)

where $V[\hat{p}_{it}] = \left(\sum_{i} u_{it} (\hat{p}_{it} - E[\hat{p}_{it}])^2\right) / \sum_{i} u_{it}$ is the unemployment-weighted variance of \hat{p}_{it} . Equation (23) is very similar to comparable expressions in Sahin, Song, Topa, and Violante (2014) and Barnichon and Figura (2015). The derivation is given in appendix B.

The amount of mismatch unemployment will depend on the level of disaggregation.

¹⁴The segment-specific separation probabilities, unlike the segment-specific job-finding probabilities, are constant, and the weights are affected less simply because $n_{it} >> u_{it}$ for almost all *i* and *t*.

At higher levels of aggregation, we would expect to see substantial mismatch within segments, so that the observed mismatch across segments is a lower bound for the actual labor market mismatch. We return to this issue in detail when we discuss our estimates for mismatch unemployment in Section 6.1.

4.4 Accounting Procedure

The decomposition of mismatch unemployment into its sources is implemented in three steps. First, we estimate empirical equivalents of the wedges $\hat{\gamma}_{it}^{WM}$, $\hat{\gamma}_{it}^{JM}$, $\hat{\gamma}_{it}^{WD}$ from equations (17), (18) and (20) using data on the worker and firm surpluses and jobfinding and vacancy-filling rates (see Section 5 below for how we obtain these data). Second, given these estimates, we use equation (21) to calculate what the distribution of job-finding probabilities over segments would be if one or more of the wedges were zero in all labor market segments. Finally, we use equations (22) and (23) to calculate the contribution of mismatch to the aggregate job-finding probability and unemployment rate.

The idea behind this procedure is that if we remove, for example, the worker-mobility frictions, setting $\hat{\gamma}_{it}^{WM}$ equal to zero, but leave the job-mobility and wage-setting frictions in place, then $\hat{\gamma}_{it}^{JM}$ and $\hat{\gamma}_{it}^{WD}$ would stay the same. A complication arises if the wedges are correlated. Then, removing a friction does not necessarily decrease mismatch and unemployment, but may increase it. This is simply because in a second-best environment, different frictions may reinforce or counteract each other, so that removing a friction may decrease efficiency.¹⁵ Potentially, the correlations between $\hat{\gamma}_{it}^{WM}$, $\hat{\gamma}_{it}^{JM}$ and $\hat{\gamma}_{it}^{WD}$ are informative about the frictions that maintain labor market mismatch. Empirically however, we find very little correlation, see Section 6.2; we therefore focus on a simple variance decomposition between the three sources of mismatch.

Any correlation between the wedges means the variance decomposition of equation (23) will depend on the order, in which we shut down the various sources of mismatch in equation (21). To understand this point, it is important to realize that the contribution of a friction is always relative to a baseline. We can estimate the contribution of a particular friction to mismatch as the difference between the aggregate job-finding rate that would prevail without this friction, leaving all other frictions as they are in the data, and the actual aggregate job-finding rate. Or, we can estimate this contribution as the difference between the aggregate that would prevail if only this friction were present, and the aggregate job-finding rate in the no-mismatch allocation. In general, the two approaches will give different answers, because the first friction that is shut down will shut down the covariance terms as well. In appendix C, we show that

¹⁵As an example, consider two otherwise identical labor market segments, one with high wages and one with low wages. Suppose these wage differentials can exist because of wage-setting frictions, but that labor market tightness is nevertheless equal in both submarkets, because mobility costs prevent workers and jobs from moving from one submarket to the other. Now suppose we were to remove the mobility costs but leave the wage-setting frictions in place. Unemployed workers would move to the submarket where wages are high, whereas vacancies would move to the submarket where wages are low. The result would be a decrease in the aggregate job-finding rate and an increase in mismatch unemployment.

the contribution of a friction that we remove includes the contribution of the covariance of that friction with other frictions in place, whereas the contribution of a friction that we introduce does not. Therefore, we calculate the contribution of each friction in both ways and average it, attributing the covariance between two frictions in equal proportions to each of the frictions. This approach, which is similar to a Shapley-Owen decomposition, guarantees that our decomposition adds up to exactly 100% of mismatch unemployment.

The most important limitation of our approach is that we needed to assume that parameter heterogeneity is time-invariant, in order to be able to identify the contribution of mismatch to unemployment. If in reality there is time variation in unemployment benefits b_{it} , vacancy maintenance costs κ_{it} and/or matching efficiency ϕ_{it} then we will incorrectly attribute the dispersion due to this heterogeneity to mismatch. Substituting out for \hat{q}_{it} , which is not directly observable, in equations (12), (13) and (15) using (14), and taking deviations, we get the following expressions for the wedges we would measure in this case.

$$\hat{p}_{it} + \hat{S}_{it}^W = \hat{\gamma}_{it}^{WM} + \hat{\alpha}_{it}^{WM} \tag{24}$$

$$\hat{S}_{it}^{F} - \frac{\mu}{1-\mu}\hat{p}_{it} = \hat{\gamma}_{it}^{JM} + \hat{\alpha}_{it}^{JM} - \hat{\alpha}_{it}^{MT}$$
(25)

$$\hat{S}_{it}^W - \hat{S}_{it}^F = \hat{\gamma}_{it}^{WD} \tag{26}$$

The wage determination wedge is not affected, but time-varying heterogeneity in unemployment benefits $\alpha_{it}^{WM} = \log (\lambda^u - b_{it})$ biases the contribution of worker mobility frictions, and time-varying heterogeneity in vacancy maintenance costs $\alpha_{it}^{JM} = \log (\lambda^v + g'(v_{it}; \kappa_{it}))$ and matching efficiency $\alpha_{it}^{MT} = (1/(1-\mu))\log \phi_{it}$ is spuriously attributed to job mobility frictions.

If we assume that the time variation in parameters is uncorrelated with the true wedges, then time-varying parameter heterogeneity will lead to an upward bias in the estimated overall amount of mismatch, and in the contribution of worker mobility and job mobility frictions, because $V\left[\hat{\gamma}_{it}^{WM} + \hat{\alpha}_{it}^{WM}\right] > V\left[\hat{\gamma}_{it}^{WM}\right]$ and $V\left[\hat{\gamma}_{it}^{JM} + \hat{\alpha}_{it}^{JM} - \hat{\alpha}_{it}^{MT}\right] > V\left[\hat{\gamma}_{it}^{JM}\right]$. If the heterogeneity in these parameters comoves with the business cycle, then the estimates for the cyclicality of total mismatch and the contribution of worker and job mobility frictions will be affected as well. Unfortunately, it is not possible to say much more about the bias we should expect without data on these parameters. However, in Section 6.4, we present estimates with additional controls for time-varying worker heterogeneity. To the extent that parameter heterogeneity is correlated with observable worker characteristics, we would expect that controlling for these characteristics would alleviate the bias due to time-varying parameter heterogeneity. We take the finding that controls for worker characteristics affect our results very little as suggestive evidence that time-varying parameter heterogeneity is not a great concern in practice.

5 Data and Measurement

To implement the mismatch accounting exercise described in the previous section, we need empirical measures of the job-finding rate p_{it} , the worker-finding rate q_{it} , worker surplus S_{it}^W , which is closely related to wages, and firm surplus S_{it}^F , closely related to profits, for submarkets of the labor market. In this section, we describe how we obtain these measures. In Section 5.1, we describe the micro-data we use to extract disaggregated measures for finding rates, wages, and profits. Then, in Sections 5.2 and 5.3, we describe how we use these data to calculate the theoretical measures we need for our accounting exercise. Here, we need to make some auxiliary assumptions, which we revisit after discussing our results in Section 6.4.

The first empirical difficulty is how to define a labor market segment or submarket. A submarket of the labor market is defined as a subset of unemployed workers or vacant jobs that are similar to each other but different from other workers or jobs, so that each unemployed worker and each firm with a vacant job searches in one submarket only. In our theoretical framework, we assumed that submarkets are mutually exclusive, so that two workers who are searching for some of the same jobs are searching for all of the same jobs, and if a worker is searching for a job, then that job is searching for that worker. In practice, these assumptions are likely to be violated, unless we define submarkets as very small and homogeneous segments of the labor markets, based on geographic location as well as the skill set required to do a job.

We use 51 U.S. states (including Washington, D.C.) to explore geographic mismatch and around 37 industries to explore skill mismatch.¹⁶ This choice is driven by data limitations and follows other empirical contributions in this literature (Şahin, Song, Topa, and Violante (2014), Barnichon and Figura (2015)). Unfortunately, it is not possible to use very small submarkets, because we would have too little data about each submarket.¹⁷ For our estimates of overall mismatch, we also use three-digit occupations, as Şahin, Song, Topa, and Violante (2014) do, which arguably better categorize similarly skilled jobs than do industry-oriented groupings; however, because data on profits by occupation are not available, we cannot use these data for our decomposition.

5.1 Data Sources

Our primary data sources are the January 1979 to December 2015 basic monthly files of the Current Population Survey (CPS) administered by the Bureau of Labor Statistics (BLS).¹⁸ We limit the sample to wage and salary workers between 16 and 65 years of

 $^{^{16}}$ We have 37 industries based on the SIC classification for the 1979-1997 period and 38 industries based on the NAICS classification for the 1998-2015 period.

 $^{^{17}}$ Shimer (2007), for instance, suggests using the interaction of 800 occupations and 922 geographic areas (362 MSAs plus 560 rural areas), which gives a total of 740,000 submarkets. In our dataset, we have information on about 150,000 workers in a given year, so that we would have 1 datapoint for each 5 submarkets.

¹⁸We mostly rely on data provided by IPUMS (Flood, King, Ruggles, and Warren (2017)), but complement these with data provided by the NBER to add the variables unemployment duration (DU-RUNEMP) before 1994 and weekly earnings (EARNWEEK) before 1989.

age, with non-missing data for state and industry classification. From the basic monthly files we construct job-finding and separation rates. We aggregate the monthly data to an annual time series in order to increase the number of observations. Our estimates of finding and separation rates are based on about 750,000 observations per year. From the outgoing rotation groups, we get wages, calculated as usual weekly earnings divided by usual weekly hours. Again, we aggregate the data to an annual time series, ending up with a sample of about 160,000 workers per year. Table 2 in appendix F lists the states and industries we use and summarizes the number of observations used to calculate the job-finding and separation rates and the average wage for the state-year and industryyear cells. The average cell size for job-finding and separation rates is 858 per year for the state-level data, and 1124 per year for the industry-level data; the smallest cells have 159 and 15 observations, respectively.

Data on profits by state and industry come from the National Income and Product Account (NIPA) data collected by the Bureau of Economic Analysis (BEA). We use gross operating surplus per employee as our measure of profits. Gross operating surplus equals value added, net of taxes and subsidies, minus compensation of employees. Net operating surplus equals gross operating surplus minus consumption of fixed capital; it is the measure of business income from the NIPA that is closest to economic profits. Because data on net operating surplus are not available at the state and industry levels, we use gross operating surplus, thus effectively assuming that fixed capital does not differ much across labor market segments. Under the assumptions of a Cobb-Douglas production technology and perfect capital markets, profits per employee equal the marginal profits from hiring an additional worker.¹⁹

In 1998, the industry classification system changed from the SIC to the NAICS. Using a consistent industry classification over the entire sample period would force us to aggregate at a higher level. Instead, we use the SIC classification until 1997, and the NAICS from 1998 onwards, using approximately the same number of industries in both subsamples. This allows us to calculate comparable cross-industry variances for \hat{p}_i , \hat{q}_i , \hat{S}_i^W , and \hat{S}_i^F over the full sample period. When adjusting for parameter heterogeneity, as described in Section 4.2, we subtract industry-specific time-series averages separately for the 1979-1997 and 1998-2015 periods.

We deflate nominal data on wages and profits using the CPI provided by the BEA (series CUUR00000SA0). Using an aggregate price deflator does not directly affect our results, because we use only the cross-sectional variation in the data, but it is important for the heterogeneity correction, which relies on subtracting the time-series mean for each segment. As a robustness check, we also show results for unemployment due to geographic mismatch using a state-specific deflator provided by Berry, Fording, and

¹⁹Let $Y = AK^{\alpha}L^{1-\alpha}$ be output, produced according to a Cobb-Douglas technology from capital K and labor L. Profits (or net operating surplus) are given by $\Pi = Y - rK - wL$, where r is the rental rate of capital, and w is the wage rate. The marginal profits from an additional employee are $d\Pi/dL = (1-\alpha)Y/L - w$, where dK/dL = 0 by the envelope theorem if capital is chosen optimally by the firm. Profits per employee are $\Pi/L = Y/L - rK/L - w$. If capital markets are frictionless, then the rental rate equals the marginal product of capital, $r = \alpha Y/K$, so that $\Pi/L = (1-\alpha)Y/L - w = d\Pi/dL$.

Hanson (2000), which is available until 2007.

Finally, we need to make assumptions about the unemployment share of the matching function μ and the discount rate r. In our baseline results, we assume $\mu = 0.72$, as in Shimer (2005), and explore the robustness of our results to setting $\mu = 0.5$ as in Sahin, Song, Topa, and Violante (2014), which is at the other end of the plausible range of estimates in Petrongolo and Pissarides (2001). We set the annual discount rate r = 0.04, but this assumption matters very little for the results.

5.2 Finding and Separation Rates

We calculate job-finding and separation rates of workers from the unemployment rates in two subsequent months and the number of newly separated workers (short-term unemployed) in the CPS. This way to measure worker flows, suggested by Shimer (2012), has the advantage that it is not subject to time-aggregation bias.²⁰ We show that our results are robust if we instead calculate job-finding and separation rates from the number of workers transitioning between unemployment and employment from matched CPS data, see Section 6.4.

Workers are attributed to the state where they live and the industry where they work. We attribute unemployed workers, for whom we lack information on industry, to the industry in which they last held a job, following standard practice at the BLS. This assumption is not consistent with our model, which would attribute unemployed workers to the industry in which they are searching for a job. We address this concern with a novel way of constructing segment-specific job finding rates, attributing unemployed workers to the industry in which they eventually find a job, see appendix G. We show in Section 6.4 that our results are robust to this alternative approach.

To calculate worker-finding rates of firms, we would need firm-level data, which are available from the Job Openings and Labor Turnover Survey (JOLTS), but only from the year 2000 onwards. To obtain data on worker-finding rates for a longer sample period, we use equation (19) to construct data for worker-finding rates of firms \hat{q}_{it} from data on job-finding rates of workers \hat{p}_{it} .

5.3 Match Surplus

As derived in appendices A.2.1 and A.2.2, the surplus of a match to the worker and the firm is given by a Bellman equation,

$$(1+r) S_{it}^{k} = E_{t} y_{it+1}^{k} + (1-\delta_{i}) E_{t} S_{it+1}^{k}$$
(27)

where $k \in \{W, F\}$ and y_{it}^k is the flow payoff from the match, which equals $w_{it} - \lambda^u$ for workers and $f(n_{it}; z_{it}) - w_{it} - \lambda^v$ for firms. We observe match payoffs y_{it}^k and separation

²⁰In February 1994, there was a change in the way unemployment duration is reported in the CPS. To correct for this change, we multiply the official short-term unemployment rate by a factor 1.1579 in months from February 1994 onwards, as suggested by Elsby, Michaels, and Solon (2009).

rates δ_i in our dataset, and use these data and equation (27) to calculate match surplus for the worker and firm, S_{it}^W and S_{it}^F respectively.²¹ In order to do this, we need to make assumptions on the evolution of these variables after a match is created.

For our exercise, what matters is the dispersion in surplus across submarkets of the labor markets. Dispersion in surplus is sensitive to the persistence in payoffs and to the segment-specific separation probabilities. The persistence of payoffs matters because match surplus equals the expected net present value of all future payoffs from the match. If payoffs are very persistent, then current payoff differentials will persist into the future, thus generating more dispersion in the expected net present value. Persistence in separation rates matters as well, because it determines to what extent the separation probabilities are segment-specific.

We assume match payoffs follow first-order autoregressive processes that revert to the average across all submarkets. The first-order autocorrelation in wages is 0.94 per year in the state-level data and 0.64 in the industry-level data based on the NAICS classification.²² This is consistent with Blanchard and Katz (1992), who find an autocorrelation of 0.94 across U.S. states, and Alvarez and Shimer (2011), who find 0.90 for 75 industries at the three-digit level of disaggregation (CES data, 1990-2008). Autocorrelation in profits is 0.96 in the state-level data and 0.72 in the industry-level data. Since wages and profits are close to a random walk in the state-level data, we use this as our baseline.²³ For industries, we assume monthly mean reversion in wages and profits of 0.037 and 0.027, respectively, consistent with the observed autocorrelations. In Section 6.4, we show that our results are robust to these assumptions.

In the model, separation rates are constant over time, and this is our baseline. However, we explore the robustness of our results to higher degrees of mean reversion, because, in the data, separation rates are quite far from a random walk: the first-order autocorrelation is 0.61 per year in the state level data and 0.50 in the industry-level data. Therefore, we allow separation rates to follow an (independent) autoregressive process as well.

Using the stochastic processes for match payoffs y_{it}^k and separation rates δ_{it} , we can solve equation (27) recursively to obtain match surplus. For convenience, we approximate around the separation rate following a random walk so that we can obtain an explicit expression for the solution, see appendix D for the derivation,

$$S_{it}^{k} \simeq \frac{\left(r + \delta_{it}\right)\left(r + \delta_{it} + \rho_{\delta}\right)}{\left(r + \delta_{it} + \rho_{\delta}\right) + \rho_{\delta}\left(1 + r + \delta_{it}\right)\left(\bar{\delta}_{t} - \delta_{it}\right)} \left(\frac{\bar{y}_{t}^{k}}{r + \delta_{it}} + \frac{y_{it}^{k} - \bar{y}_{t}^{k}}{r + \delta_{it} + \rho_{y}^{k}}\right)$$
(28)

²¹We also need values for the parameters λ^{u} and λ^{v} . We assume $\lambda^{v} = 0$ to reflect free entry of vacancies. We also set $\lambda^{u} = 0$ for symmetry, but explore the robustness of our results to this assumption in Section 6.4.

 $^{^{22}}$ We report simple first-order autocorrelations in this paragraph. However, the persistence in the data is very similar and, if anything, higher if we use the coefficient on the lagged dependent variable in a dynamic panel regression with fixed-effects.

²³Strictly speaking, what matters is not the persistence in average wages and profits, but the persistence of wages and profits of a given match. However, as shown by Haefke, Sonntag, and van Rens (2013) and Kudlyak (2014), wages paid out over the duration of a match are more persistent than average wages, so, these estimates, if anything, understate the autocorrelation in wages.

where \bar{y}_t^k is the average of y_{it}^k across segments *i*. In this expression, ρ_{δ} and ρ_y^k denote the mean reversion in separation probabilities and match payoffs, so that the autocorrelation coefficient in these variables equals $1 - \rho_{\delta}$ and $1 - \rho_y^k$, respectively. The approximation is good if separation rates are persistent, i.e. for ρ_{δ} close to zero.

If match payoffs follow a random walk, $\rho_y^k = 0$, and turnover is constant, $\rho_{\delta} = 0$, as in our baseline, then match surplus is the annuity value of the current payoff, $S_{it}^k = y_{it}^k/(r + \delta_{it})$, evaluated at an effective discount rate which includes not only the rate of time preference, but also the separation probability. The higher the wage in a submarket, the higher is the surplus of having a job in that submarket. The more likely it is to lose that job in the future – that is, the higher is δ_{it} – the lower is the surplus.

6 Results

We start the description of our results with our estimates for the total effect of mismatch on the level and the cyclicality of unemployment. Then, we turn to the decomposition of mismatch into its three sources. We explore how well no-mismatch equilibrium conditions (7), (8) and (10) hold in the data in Section 6.2, and we present the results of our mismatch accounting exercise in Section 6.3.

6.1 Mismatch Unemployment

Figure 1 plots the unemployment rate that is due to mismatch across occupations, across states (panel A) and across industries (panel B) over the 1979-2015 period. For comparison, the graphs also show the overall unemployment rate over the same period, although on a different scale. We will use the series in these graphs to address the questions of how large the impact of labor market mismatch is on unemployment, and how it fluctuates over the business cycle.

Unemployment due to mismatch across three-digit occupation-state cells averages almost 1 percentage point. Comparing to an average unemployment rate of around 7%, it is clear that occupational mismatch is responsible for a substantial part (around 13%) of unemployment. Mismatch across states and industries is much smaller at 0.05 to 0.1 percentage points, or 0.7 to 1.5% of unemployment, respectively. Clearly the level of disaggregation matters for the observed amount of mismatch, and there is likely to be substantial mismatch within states and within industries. However, the cyclical pattern of mismatch unemployment looks similar across occupations, states, industries. It is worthwhile, therefore, to further explore mismatch across states and industries, as a proxy for overall labor market mismatch.²⁴

Figure 1 clearly shows that the cyclical fluctuations in mismatch unemployment are very similar to those in the overall unemployment rate. Mismatch unemployment

²⁴An additional argument why mismatch across industries and states may be a reasonable proxy for mismatch across occupations is described in appendix E, where we use a back-of-the-envelope correction for aggregation to argue that the estimates for mismatch across states and industries imply a level of mismatch that is of a similar magnitude as the observed mismatch across occupations.

closely follows the business cycle, rising in the 1982, 1991, 2001 and 2008 recessions, and declining slowly during the recoveries, as does the unemployment rate. The relative amplitude of these fluctuations is very similar to those in the total unemployment rate. There is no evidence that mismatch unemployment is less cyclical or more persistent than the overall unemployment rate. Finally, there is no indication that the fraction of the increase in unemployment in the Great Recession that was due to mismatch was larger than in other recessions.

To obtain a summary statistic for the importance of mismatch to the overall unemployment rate, we regress mismatch unemployment u_t^{MM} on a constant and the overall unemployment rate u_t in deviation from its average \bar{u} .

$$u_t^{MM} = \beta_0 \bar{u} + \beta_1 \left(u_t - \bar{u} \right) \tag{29}$$

The intercept β_0 in this regression measures the contribution of mismatch to the average level of unemployment, whereas the slope coefficient β_1 measures the contribution of mismatch to fluctuations in unemployment. We report both statistics in Table 1, see the rows labelled "Mismatch across occupations" and "Baseline." The contribution of mismatch to fluctuations in unemployment is roughly similar to the contribution to the level of unemployment: 20% for mismatch across occupations, and 3.2% and 0.7% for mismatch across states and industries, respectively.

Figure 2 shows an alternative presentation of these results on the contribution of mismatch to unemployment for mismatch across states (panel A) and industries (panel B). Here, we plot $\log \bar{u}_t - \log \bar{u}_t^* \simeq (\bar{u}_t - \bar{u}_t^*) / \bar{u}_t$, which measures unemployment due to mismatch as a fraction of the overall unemployment rate. The four lines in these graphs represent the exact contribution of mismatch to unemployment, calculated using equations (51) and (53) in appendix B, and the approximation using equation (23), with and without controlling for parameter heterogeneity. It is clear from these graphs that the log-normal approximation used to derive equation (23) is very good. It is also clear that controlling for parameter heterogeneity is important. Without these controls, the contribution of mismatch to unemployment is about two (for states) to three (for industries) times larger. Moreover, the effect of parameter heterogeneity changes over time and with the business cycle. Of course, as explained in Section 4.2, it is possible that by taking out parameter heterogeneity, we are also removing some inefficient dispersion in job-finding rates. Therefore, our estimates for mismatch unemployment should be thought of as a lower bound. There is no obvious cyclical pattern in the graphs of the relative contribution of mismatch to unemployment, confirming our finding that mismatch comoves with the overall unemployment rate.

Our estimates are broadly in line with Şahin, Song, Topa, and Violante (2014), who employ different data and conclude that geographic mismatch is very small, but industry-level mismatch (at the two-digit level) explains around 14% of the increase in unemployment in the Great Recession. The estimates in their Figure 3 imply a similar contribution of mismatch to the level of unemployment (although the authors do not

report this in their text). Consistent with our argument that aggregation importantly biases the estimate of the contribution of mismatch, Şahin, Song, Topa, and Violante (2014) also find that when they disaggregate further, to three-digit occupation level, the contribution of mismatch increases to 29%. However, we emphasize that our estimates of the contribution of mismatch to the level of unemployment are rough, and the estimates in Şahin, Song, Topa, and Violante (2014) are the more credible ones. The reason is that while we use data on worker flows from the CPS, Şahin, Song, Topa, and Violante (2014) also use disaggregated data on vacancies from JOLTS and HWOL, which are available only from December 2000 and May 2005, respectively. The contribution of the current study is in the estimates of the sources of mismatch, to which we now turn.

6.2 Deviations from the No-Mismatch Conditions

We now turn to the most interesting part of our results: the decomposition of mismatch unemployment into the sources of the mismatch. The idea behind our accounting exercise is to compare the relationship between job-finding rates and worker surplus, between job-filling rates and firm surplus, and between worker and firm surpluses to the predictions of the model for these relationships in the absence of mismatch. We will start, therefore, by exploring what these relationships look like in the data.

Figure 3 shows scatterplots of states around the worker-mobility, job-mobility and wage-determination conditions. These graphs are for 2010, but they look similar for other years. Deviations across states from the no-mismatch conditions are large and systematic. Nevertheless, the positive correlation between worker and firm surpluses predicted by wage-setting condition (10), and the negative correlation between firm surplus and job-filling rates predicted by the worker-mobility condition (7) are somewhat visible in the data. These graphs suggest that frictions preventing firms from moving jobs across states play the most important role as a source of mismatch. Figure 4 shows similar results for the equilibrium conditions across industries. The plots look similar to those for states, and again suggest that job-mobility frictions are the most important source of mismatch.

Scatter plots around the no-mismatch equilibrium conditions may not give the full picture if the deviations from these conditions are correlated. However, in practice this is not much of a concern. The correlation between the wedges in the worker-mobility condition $\hat{\gamma}_{it}^{WM}$ and the wedges in the job-mobility condition $\hat{\gamma}_{it}^{JM}$ is on average -0.23 for states, 0.07 for (SIC) industries over the 1979-1997 period, and -0.14 for (NAICS) industries 1998-2015. Given these low correlations, it makes sense to think of barriers to worker and job mobility as separate frictions.

6.3 Sources of Mismatch

Figure 5 shows the results of our mismatch accounting exercise described in Section 4. The figure shows the evolution over time of the three sources of unemployment due to mismatch across states and industries.

Mismatch across states is mainly generated by job-mobility frictions, with a smaller contribution of worker-mobility frictions and virtually no role for wage-setting frictions. Mismatch across industries looks very similar, with a possibly even smaller role for worker-mobility frictions.

To summarize the contribution of each source of mismatch to the unemployment rate, we regress mismatch unemployment due to each source on the total unemployment rate due to mismatch (in deviation from its mean).

$$u_t^{XX} = \beta_0^{XX} \bar{u}^{MM} + \beta_1^{XX} \left(u_t^{MM} - \bar{u}^{MM} \right)$$
(30)

where XX stands for the source of mismatch, i.e., $XX \in \{WM, JM, WD\}$.²⁵ The intercept in this regression measures the contribution of each of the frictions to the average level of mismatch unemployment, so that $\beta_0^{XX} = \bar{u}^{XX}/\bar{u}^{MM}$, whereas the slope coefficient measures the contribution of mismatch to fluctuations in unemployment. The slope coefficient captures both the degree of correlation of unemployment due to a particular source of mismatch with the total mismatch unemployment rate and the size of fluctuations in mismatch due to that source, i.e., $\beta_1^{XX} = corr(u_t^{XX}, u_t^{MM}) sd(u_t^{XX}) / sd(u_t^{MM})$. Note that because $u_t^{WM} + u_t^{JM} + u_t^{WD} = u_t^{MM}$, the contributions of the three sources to the total add up to one, i.e. $\beta_0^{WM} + \beta_0^{JM} + \beta_0^{WD} = 1$ and $\beta_1^{WM} + \beta_1^{JM} + \beta_1^{WD} = 1$, so that this is an exact decomposition.

Results of the mismatch accounting exercise are reported in Table 1. Barriers to job mobility contribute 79% to the level of and 80% to the fluctuations in mismatch across states and 78% to the level and 65% to the fluctuations of mismatch across industries. Barriers to worker mobility are responsible for roughly a fourth of that: 20% of the level and 19% of the fluctuations of mismatch across states, and 19% of the level and 27% of the fluctuations of mismatch across industries. The contribution of wage-setting frictions to mismatch is virtually zero, both for the level of mismatch and for its fluctuations, and both across states and across industries.

6.4 Robustness

A number of assumptions were necessary to construct the data needed for our analysis. In this subsection we explore the robustness of our results to these assumptions. We summarize the results in terms of the contribution of mismatch to the level and fluctuations of the overall unemployment rate, as explained in Section 6.1; and in terms of the contribution of barriers to worker mobility, barriers to job mobility and deviations from the efficient wage determination equation to labor market mismatch, as described in Section 6.3. These summary statistics are presented for a number of robustness checks in Table $1.^{26}$ In the top and bottom panels of this table, the first line shows our baseline

 $[\]overline{{}^{25}u_t^{MM} = (\log \bar{u}_t - \log \bar{u}_t^*) \bar{u}_t}$, where $\log \bar{u}_t - \log \bar{u}_t^*$ as in (22) and (23), and u_t^{XX} is calculated similarly. ²⁶A few entries in the table are negative. The contribution of a friction to mismatch unemployment may be negative because the correlation between deviations from the different no-mismatch conditions

estimates for state-level and industry-level data, respectively.

For the construction of job-filling rates from job-finding rates, we made the assumption that the matching technology is well described by a Cobb-Douglas matching function with an elasticity of unemployment μ of 0.72, see Sections 3.3 and 5.2. In the table, we show the effect of varying this elasticity from 0.5 or 0.8. A higher (lower) elasticity increases (decreases) the concavity of the aggregate job-finding rate in the segment-specific job-finding rates, see Section 4, and therefore increases (decreases) the estimated contribution of mismatch to unemployment. This effect is fairly strong for the estimates for the overall amount of mismatch, but the decomposition into its sources is largely unaffected.

We used short-term unemployment to measure job-finding and separation rates, as suggested by Shimer (2012). Our results are very similar if we instead measure worker flows from transitions of workers between unemployment and employment using basic monthly CPS data matched between subsequent months. In what is possibly the most contentious assumption we made in measuring worker flows, we calculated job-finding rates by industry of origin rather than by industry of destination. It is not possible to calculate these rates in the same way by industry of destination, because we do not have information about in which industry unemployed workers are searching for a job. However, we can use matched CPS data, combined with information on unemployment duration, to back out a rough measure of job-finding rates, see appendix G. This measure is much noisier than our baseline measure, because it uses a smaller sample consisting only of unemployed workers who find a job within the sample. Nevertheless, the results are very similar.

For the *construction of match surpluses*, we made a number of choices, among which the assumptions that price deflators are the same across states, see Section 5.1, that the opportunity cost of an additional unemployed worker λ^u equals zero, that match payoffs (wages or profits) follow an autoregressive process, and that match turnover is constant, see Section 5.3. The next set of lines in the table explores the robustness of our results to these assumptions. Since none of these assumptions affect the observed dispersion in job-finding rates, the estimates of the contribution of mismatch to unemployment are not affected, except for the state-specific price deflators, which generate slightly more dispersion and, therefore, a slightly higher estimate of mismatch unemployment.²⁷ The composition of mismatch into its sources is affected, but the effect is very small. To explore the effect of non-zero opportunity costs of unemployment, we vary the home production of non-participants λ^u from 0 to 20, 50 and 70% of the global average wage. For $\lambda^u \geq 50\%$, this leads to negative worker surplus in some industries in some year. When this happens, we drop the entire industry from the sample, which explains why also the contribution of overall mismatch changes in these cases. Despite the changing sample, however, the results for both the overall level of mismatch and its composition

is not exactly zero.

²⁷We use a state-specific deflator provided by Berry, Fording, and Hanson (2000), which is available until 2007.

change very little.

For some assumptions, we cannot directly explore robustness, but we can argue they are unlikely to affect our findings. Measurement error, while substantial, clearly does not drive our results. Classical measurement error would generate non-systematic deviations from all equilibrium relationships, whereas we find the wage-setting conditions, and, to a lesser extent, the worker-mobility condition in the data, while deviations from all three conditions look systematic. A similar argument can be made for on-the-job search. If workers are searching for a new job while employed, this increases workers' match surplus, but, given observed wages and job-finding rates, this effect is not taken into account in the way we construct match surplus, see Section 5.3. If on-the-job search intensity is the same for all workers and all firms, then this does not affect our results, because we work in deviations from the cross-sectional mean. If on-thejob search intensity varies systematically with the value of a match, then on-the-job search would increase or decrease the slope of the worker-mobility curve, (mis)leading us to conclude that worker-mobility frictions are giving rise to mismatch. For the same reason, our findings cannot be driven by workers looking for jobs in surrounding regions and occupations as Marinescu and Rathelot (2016) show they do, because this effect would also make it less likely to find a worker-mobility condition in the data.

A more serious issue is that of *discouraged workers*. It is possible that unemployed workers leave labor market segments with low surplus (wages), not by moving to a different labor market segment, but by dropping out of the labor force. This mechanism would make it seem like the no-arbitrage condition for worker mobility is satisfied, while there is substantial mismatch, leading not to unemployment but to non-employment. Unfortunately, without better data, there is very little we can do to explore this issue.

Finally, we explore the effect of *heterogeneity*. As described in Section 4.2, in our baseline we control for time-invariant heterogeneity by removing the segment-specific time-series averages from \hat{p}_{it} , \hat{S}_{it}^W , \hat{q}_{it} and \hat{S}_{it}^F , similar to controlling for fixed effects in a regression. Next, we explore the robustness of our results if we control for observed worker heterogeneity. If demographic characteristics of the workforce change over time, removing time-series averages does not control for this effect. We address this concern by adjusting worker- and job-finding rates, separation rates and earnings for observable worker characteristics, by regressing earnings as well as the likelihood of becoming unemployed and experiencing short-term unemployment on potential labor market experience (age minus years of schooling minus 6), and dummies for educational attainment, gender, race, marital status.²⁸ For each year and for each state or industry, we then impute the fitted values for a hypothetical worker with average demographics. When we do this, our results change very little.

 $^{^{28}}$ For wages we use a log-linear specification, as is common in the literature, see Card (1999). In order to get fitted values for wages, we use the fitted values for log wages and apply the correction factor suggested by Cameron and Trivedi (2010).

7 Conclusions

Mismatch unemployment is unemployment due to inefficient dispersion in labor market conditions across submarkets. We proposed an accounting framework using two arbitrage equations and an efficient wage determination equation that allows us to estimate mismatch unemployment and decompose it into its sources. This framework uses data on the values of unemployment and vacancies, rather than on their quantities, as inputs; thus, available data allowed us to present estimates for the 1979-2009 period, going further back in time than previous studies. More importantly, this paper is the first to report on the causes of mismatch.

We argue that mismatch was quantitatively important in the United States over the last four decades, with mismatch across detailed occupations explaining around 13% of unemployment. The cyclical behavior of mismatch unemployment is very similar to that of the overall unemployment rate. This finding is driven by the fact that dispersion in labor market conditions across states and industries moves closely with the business cycle. The unemployment that derives from this dispersion is as cyclical as the overall unemployment rate, and it is no more persistent. As a corollary, the nature of the increase in unemployment in the Great Recession is no different from previous recessions, although it is of course more severe.

The underlying frictions that causes mismatch to persist are predominantly barriers to job mobility, which explains between 65% and 80% of mismatch across both states and industries. The main reason for these frictions is probably the production technology, which allows only for limited possibilities for substituting one type of worker for another. Barriers to entry into particular industries, or lack of industry-specific know-how or brand recognition may also play a role.

We find a much smaller role for worker-mobility frictions, which account only for about 20% of mismatch. This finding is perhaps surprising in light of the debate on policies aimed at increasing worker mobility, see e.g., Katz (2010). On the other hand, the finding is consistent with the observation that U.S. workers are quite flexible, and are willing, for instance, to move between states in order to find a job (Molloy, Smith, Trezzi, and Wozniak (2016)).

Wage-setting frictions play virtually no role for mismatch. This finding is consistent with the flexibility of the U.S. market, which is characterized by flexible wages, particularly for newly hired workers (Haefke, Sonntag, and van Rens (2013)), and by low levels of unionization since the 1980s (Farber and Western (2002)). We make no claim that our results apply to different countries or different time periods, and it is quite possible that in European countries, for instance, both worker-mobility frictions and wage-setting frictions are much more important due to language barriers and more rigid wage-setting institutions.

References

- Abraham, Katharine. (1987). Help-wanted advertising, job vacancies, and unemployment. Brookings Papers on Economic Activity 18(1), 207–248.
- Abraham, Katharine G and Katz, Lawrence F. (1986). Cyclical unemployment: Sectoral shifts or aggregate disturbances? Journal of Political Economy 94(3), 507– 22.
- Alvarez, Fernando and Robert Shimer. (2011). Search and rest unemployment. Econometrica 79(1), 75–122.
- Barnichon, Regis and Andrew Figura. (2010). What drives movements in the unemployment rate? a decomposition of the Beveridge curve. Technical report.

(2015). Labor Market Heterogeneity and the Aggregate Matching Function. American Economic Journal: Macroeconomics 7(4), 222–249.

- Berry, William D. and Richard C. Fording and Russell L. Hanson. (2000). An annual cost of living index for the American states, 1960-95. *Journal of Politics* 60(2), 550–67.
- Birchenall, Javier A. (2011). A Competitive Theory of Mismatch. mimeo, University of California at Santa Barbara.
- Blanchard, Olivier and Peter Diamond. (1989). The Beveridge curve. Brookings Papers on Economic Activity 20(1), 1–76.
- Blanchard, Olivier Jean and Lawrence F. Katz. (1992). Regional evolutions. Brookings Papers on Economic Activity 23(1), 1–76.
- Cameron, A. Colin and Pravin K. Trivedi. (2010). Microeconometrics Using Stata (Revised ed.). Stata Press books. StataCorp LP.
- Card, David. (1999). The causal effect of education on earnings. In O. Ashenfelter and D. Card (Eds.), *Handbook of Labor Economics*, Volume 3A, pp. 1801–1863. Elsevier.
- Carrillo-Tudela, Carlos and Visschers, Ludo. (2013, January). Unemployment and Endogenous Reallocation over the Business Cycle. IZA Discussion Papers 7124, Institute for the Study of Labor (IZA).
- Chang, Briana. (2011). A Search Theory of Sectoral Reallocation. Department of finance, investment and banking working papers, University of Wisconsin, Madison.
- Chari, V. V. and Kehoe, Patrick J. and McGrattan, Ellen R. (2007). Business cycle accounting. *Econometrica* 75(3), 781–836.
- Coles, Melvyn G. and Jones, Paul and Smith, Eric. (2010). A picture of stock-flow unemployment in the united kingdom. *Macroeconomic Dynamics* 14(04), 427–453.
- DeLong, Bradford. (2010). The sad thing is that Narayana Kocherlakota was supposed to be the smart one among the Minnesota economists... September 26, http://delong.typepad.com/sdj/2010/09/the-sad-thing-is-thatnarayana-kocherlakota-was-supposed-to-be-the-smart-one-among-the-minnesotaeconomists.html.
- Diamond, Peter. (1982). Aggregate demand management in search equilibrium. Journal of Political Economy 90(5), 881–94.
 - (2011). Unemployment, vacancies, wages. American Economic Review 101(4), 1045–72.

- Elsby, Michael W. L. and Bart Hobijn and Aysegül Şahin. (2010). The labor market in the Great Recession. *Brookings Papers on Economic Activity* 41(1), 1–69.
- Elsby, Michael W. L. and Michaels, Ryan and Solon, Gary. (2009). The ins and outs of cyclical unemployment. *American Economic Journal: Macroeconomics* 1(1), 84–110.
- Estevão, Marcello M. and Evridiki Tsounta. (2011, May). Has the Great Recession Raised U.S. Structural Unemployment? IMF Working Papers 11/105, International Monetary Fund.
- Farber, Henry S. and Bruce Western. (2002). Ronald Reagan and the politics of declining union organization. British Journal of Industrial Relations 40(3).
- Flood, Sarah and Miriam King and Steven Ruggles and J. Robert Warren. (2017). Integrated Public Use Microdata Series, Current Population Survey: Version 5.0. [dataset]. Technical report, University of Minnesota.
- Frey, William H. (2009). The Great American Migration Slowdown: Regional and Metropolitan Dimensions. Report, Brooking Institution.
- Furlanetto, Francesco and Nicolas Groshenny. (2016). Mismatch Shocks and Unemployment During the Great Recession. Journal of Applied Econometrics 31(7), 1197–1214.
- Groshen, Erica L. and Simon Potter. (2003). Has structural change contributed to a jobless recovery? Current Issues in Economics and Finance, Federal Reserve Bank of New York 9(8), 1–48.
- Haefke, Christian and Sonntag, Marcus and van Rens, Thijs. (2013). Wage rigidity and job creation. Journal of Monetary Economics 60(8), 887–899.
- Harris, John R. and Todaro, Michael P. (1970). Migration, unemployment and development: A two-sector analysis. *The American Economic Review* 60(1), pp. 126–142.
- Herz, Benedikt. (2017). Specific Human Capital and Wait Unemployment. Journal of Labor Economics forthcoming.
- Hosios, Arthur J. (1990). On The Efficiency of Matching and Related Models of Search and Unemployment. *Review of Economic Studies* 57(2), 279–298.
- Katz, Lawrence. (2010). Long term unemployment in the Great Recession. Testimony for the Joint Economic Committee, U.S. Congress, April 29.
- Kocherlakota, Narayana. (2010). Inside the FOMC. Speech in Marquette, MI, on August 17 as president of the Federal Reserve Bank of Minneapolis.
- Krugman, Paul. (2010). Structure of excuses. September 27, New York Times.
- Kudlyak, Marianna. (2014). The cyclicality of the user cost of labor. Technical Report C.
- Lilien, David M. (1982). Sectoral shifts and cyclical unemployment. Journal of Political Economy 90(4), 777–93.
- Lipsey, Richard G. (1965). Structural and deficient-demand unemployment reconsidered. In A. M. Ross (Ed.), *Employment Policy and the Labor Market*, pp. 210–255. UC Berkeley Press.
- Lubik, Thomas A. (2013). The shifting and twisting Beveridge curve: an aggregate perspective. Technical report.

- Marinescu, Ioana and Roland Rathelot. (2016). Mismatch unemployment and the geography of job search. American Economic Journal: Macroeconomics forthcoming.
- Molloy, Raven and Christopher L. Smith and Riccardo Trezzi and Abigail Wozniak. (2016). Understanding Declining Fluidity in the U.S. Labor Market. Brookings Papers on Economic Activity 47(1 (Spring), 183–259.
- Montgomery, James D. (1991, February). Equilibrium Wage Dispersion and Interindustry Wage Differentials. The Quarterly Journal of Economics 106(1), 163– 79.
- Mortensen, Dale T. (1982). Property rights and efficiency in mating, racing, and related games. *American Economic Review* 72(5), 968–79.
- New York Times. (2014). A Fed Policy Maker, Changing His Mind, Urges More Stimulus. Technical report, http://nyti.ms/19XrUVw.
- Lazear. Edward P. and James R. Spletzer. (2012). The United States labor market: status quo or a new normal? *Proceedings - Economic Policy Symposium - Jackson Hole*, 405–451.
- Petrongolo, Barbara and Christopher A. Pissarides. (2001). Looking into the black box: A survey of the matching function. *Journal of Economic Literature* 39(2), 390–431.
- Pilossoph, Laura. (2014). Sectoral Shocks and Move Unemployment. Technical report.
- Pissarides, Christopher A. (1985). Short-run equilibrium dynamics of unemployment vacancies, and real wages. *American Economic Review* 75(4), 676–90.
- Roback, Jennifer. (1982). Wages, rents, and the quality of life. Journal of Political Economy 90(6), 1257–78.
- Rosen, Sherwin. (1979). Wages-based indexes of urban quality of life. In P. Mieszkowski and M. Straszheim (Eds.), *Current Issues in Urban Economics*. Johns Hopkins University Press.
- Şahin, Aysegül and Song, Joseph and Topa, Giorgio and Violante, Giovanni L. (2014). Mismatch unemployment. American Economic Review 104 (11), 3529–64.
- Shimer, Robert. (2005). The cyclical behavior of equilibrium unemployment and vacancies. American Economic Review 95(1), 25–49.
 - (2007). Mismatch. American Economic Review 97(4), 1074–1101.
- Shimer, Rober. (2012). Reassessing the ins and outs of unemployment. Review of Economic Dynamics 15(2), 127–148.
- Wiczer, David. (2013). Long-term Unemployment: Attached and Mismatched? Technical report.
- Wong, Jacob. (2012, July). Aggregate Reallocation Shocks and the Dynamics of Occupational Mobility and Wage Inequality. School of Economics Working Papers 2012-04, University of Adelaide, School of Economics.

Table 1.A							
Robustness	Analysis,	mismatch	across	states			

	M	MU	Sources of MMU					
		cycle	level		cycle			
Mismatch across occupations	13.3	20.0						
Mismatch across states			WM	JM	WD	WM	JM	WD
Baseline	1.5	3.2	20	79	1	19	80	1
Elasticity matching function, $\mu = 0.5$	0.6	1.3	36	62	2	34	64	2
, $\mu = 0.6$	0.9	1.9	29	70	2	27	71	2
, $\mu = 0.7$	1.4	2.9	22	77	1	20	78	1
$__ \mu = 0.8$	2.3	4.9	14	85	1	14	86	1
Finding rates using transitions	1.6	2.8	34	64	1	33	65	2
State-specific deflators	1.3	2.8	23	78	-1	26	72	1
Opportunity cost job seeking, $\lambda^u = 0.2\bar{w}$	1.5	3.2	20	79	2	18	80	2
$____, \lambda^u = 0.5\bar{w}$	1.5	3.2	18	79	2	15	80	5
$____, \lambda^u = 0.7\bar{w}$	1.5	3.2	16	79	6	8	80	12
Mean-reversion payoffs, $\delta_y = 0$	1.5	3.2	20	79	1	19	80	1
, $\delta_y = 0.02$	1.5	3.2	21	78	1	20	79	1
, $\delta_y = 0.04$	1.5	3.2	21	78	1	21	78	1
, $\delta_y = 0.06$	1.5	3.2	21	78	0	22	78	1
, $\delta_y = 0.08$	1.5	3.2	21	78	0	22	78	1
Mean-reversion separation rate, $\delta_{\tau} = 0$	1.5	3.2	20	79	1	19	80	1
, $\delta_{\tau} = 0.02$	1.5	3.2	23	76	1	21	78	1
, $\delta_{\tau} = 0.04$	1.5	3.2	24	75	1	22	77	1
, $\delta_{\tau} = 0.06$	1.5	3.2	24	74	1	23	76	1
, $\delta_{\tau} = 0.08$	1.5	3.2	25	74	1	23	76	1
Control for observed worker heterogeneity	1.4	2.9	25	75	1	24	75	1

Table 1.B							
Robustness	Analysis,	$\operatorname{mismatch}$	across	$\operatorname{industries}$			

	MI	MU	Sources of MMU					
		cycle	level			cycle		
Mismatch across occupations	13.3	20.0						
Mismatch across industries			WM	$_{\rm JM}$	WD	WM	JM	WD
Baseline	0.7	0.7	19	78	4	27	65	8
Elasticity matching function, $\mu = 0.5$	0.3	0.3	33	60	7	49	37	14
, $\mu = 0.6$	0.4	0.4	27	68	5	39	49	11
, $\mu = 0.7$	0.6	0.6	20	76	4	29	62	9
, $\mu = 0.8$	1.1	1.1	13	84	3	20	75	6
Finding rates using transitions	0.8	0.5	27	71	1	30	65	5
Finding rates by destination	0.9	1.0	26	75	-1	24	75	1
Opportunity cost job seeking, $\lambda^u = 0.2\bar{w}$	0.7	0.7	18	78	4	27	65	8
$__ \lambda^u = 0.5\bar{w}$	0.8	0.9	17	78	5	28	64	8
$__ \lambda^u = 0.7\bar{w}$	0.9	0.8	17	78	5	24	71	5
Mean-reversion payoffs, $\delta_y = 0.02$	0.7	0.7	18	78	4	27	64	9
, $\delta_y = 0.04$	0.7	0.7	19	78	3	27	66	7
, $\delta_y = 0.06$	0.7	0.7	19	78	3	28	67	5
, $\delta_y = 0.08$	0.7	0.7	19	79	2	28	68	4
Mean-reversion separation rate, $\delta_{\tau} = 0$	0.7	0.7	19	78	4	27	65	8
, $\delta_{\tau} = 0.02$	0.7	0.7	24	72	4	29	63	8
, $\delta_{\tau} = 0.04$	0.7	0.7	26	71	4	29	63	8
, $\delta_{\tau} = 0.06$	0.7	0.7	26	70	4	28	64	8
, $\delta_{ au} = 0.08$	0.7	0.7	26	70	4	28	64	8
Control for observed worker heterogeneity	0.6	0.8	21	73	6	28	58	15

Figure 1 Mismatch unemployment



A. Across US states

B. Across industries



Unemployment due to mismatch across occupations, states and industries, calculated as explained in Section 4.3. The dashed line shows the actual unemployment rate for comparison.

Figure 2 Unemployment due to mismatch as a percentage of total unemployment



A. Across US states

B. Across industries



Percentage increase in the unemployment rate due to mismatch, with and without controlling for heterogeneity, and approximation as explained in Section 4.

Figure 3 Worker mobility, job mobility and wage determination curves across US states



A. Worker mobility condition

B. Job mobility condition



C. Wage setting condition



Lines represent the equilibrium relations corresponding to a labor market without any mismatch. Data are for 2010.

Figure 4 Worker mobility, job mobility and wage determination curves across industries



A. Worker mobility condition





C. Wage setting condition



Lines represent the equilibrium relations corresponding to a labor market without any mismatch. Data are for 2010.

Figure 5 Sources of labor market mismatch



A. Across US states

B. Across industries



The thicker line is our baseline estimate for the aggregate effect of mismatch as in Figure 2. The other lines show the contribution of worker mobility costs (WM), job mobility costs (JM) and wage setting frictions (WD) to mismatch, see Section 4.